\begin{table}%t1
\caption{\label{eliminated}Sources eliminated from the SIMBAD's pre-main-sequence list.}  
%\centering
\par
\begin{tabular}{l l r}
\hline\hline \noalign{\smallskip}
Name & Adopted object type & Reference \\ 
\hline
VY Ari       & Late type stars in binary system & 1 \\   
V573 Per  & G8 subgiant  & 2 \\
HD 21051 & K0 giant  & 2  \\
UX Ari       & Late type stars in binary system & 3 \\
46 Per      & Emission line OB stars & 4 \\
V491 Per  & G8 subgiant & 2 \\
V582 Per  & F7 dwarf & 5 \\
V492 Per  & K1 giant in binary system & 6 \\
111 Tau    & F8 dwarf & 7 \\
HD 217086 & O7 dwarf & 8 \\
\hline
\end{tabular}
\tablebib {1: Biazzo et~al. (\cite{biazzo2006}); 2: De Mediros \& Mayor (\cite{medeiros1999}); 3:~Rosario et~al. (\cite{rosario2007}); 4: Puls et~al. (\cite{puls2006}); 5: Busa et~al. (\cite{busa2007}), 6: \mbox{De Medeiros} et~al. (\cite{medeiros2002}); 7: Cenarro et~al. (\cite{cenarro2007}); 8: Mokiem et~al. (\cite{mokiem2005}).}
\end{table}